Author’s response to reviews

Title: Comparative study of ultrasound-guided paravertebral block versus intravenous tramadol for postoperative pain control in percutaneous nephrolithotomy

Authors:

Zehra Hatipoğlu (hatipogluzehra@gmail.com)
Ersel Gulec (gulecersel@yahoo.com)
Mediha Turktan (mediturktan@gmail.com)
Volkan Izol (volkanizol@yahoo.com)
Atilla Aridoğan (aaridogan@yahoo.com)
Yasemin Gunes (ygunes@cu.edu.tr)
Dilek Ozcengiz (dilekozcengiz@gmail.com)

Version: 3 Date: 22 Jan 2018

Author’s response to reviews:

Editor Comments:

1. Please include the trial registration number (TRN) and date of registration as the last line of the manuscript abstract. Please also state ‘retrospectively registered’ if the registration was made after the first participant is recruited.

The trial registration should take the following form:

Trial registration: name of registry, trial registration number, date of registration.

2. Ethics approval and consent to participate

-Please rename the section heading 'Ethics approval' to 'Ethics approval and consent to participate'

-For Consent to Participate, please state clearly whether written informed consent was obtained from the participants in this section.

3. Consent for publication
Please include a 'Consent for publication' section in Declarations and state Not Applicable as your manuscript doesn't contain any individual identifying data

4. Availability of data and materials

It is stated that 'All data generated or analyzed during this study are including in this published article.' However, this statement is not valid. This is because Availability of Data and Materials refers to the raw data-set upon which the summary statistics in the manuscript are based.

For your information: Here are some examples of Availability of Data and Materials statements that you may choose to adapt if you wish:

- The datasets generated and analysed during the current study are available in the [NAME] repository, [PERSISTENT WEB LINK TO DATASETS]
- The datasets generated and analysed during the current study are available from the corresponding author on reasonable request.
- The datasets generated and analysed during the current study are not publicly available due [REASON WHY DATA ARE NOT PUBLIC] but are available from the corresponding author on reasonable request.
- The datasets generated and analysed during the current study are not publicly available due [REASON WHY DATA ARE NOT PUBLIC].

5. Upon resubmission, please make sure that your manuscript contains the following seven sections under the heading 'Declarations' at the end of the main text:

(1) Ethics approval and consent to participate
(2) Consent for publication
(3) Availability of data and material
(4) Competing interests
(5) Funding
(6) Authors' contributions
(7) Acknowledgements

6. Please place the 'Declarations' section to be after the main text and before the References

7. Figures
• Figure titles (max 15 words) and legends (max 300 words) should be provided in the main manuscript, not in the graphic file. Please remove the figure titles from the graphic file.

8. Upon resubmission, Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Please ensure the document is in the final form for publication; please upload only files that are to be published. If you wish to respond to these revision requests, please insert the information in to the designated input box only.

Response to Editor

We thank to Editor for valuable comments, and we have added your significant comments in the relevant sections.